Alternative cascade-testing protocols for identifying and managing patients with familial hypercholesterolaemia: systematic reviews, qualitative study and cost-effectiveness analysis

Nadeem Qureshi,^{1*} Bethan Woods,² Rita Neves de Faria,² Pedro Saramago Goncalves,² Edward Cox,² Jo Leonardi-Bee,¹ Laura Condon,¹ Stephen Weng,³ Ralph K Akyea,¹ Barbara Iyen,¹ Paul Roderick,⁴ Steve E Humphries,⁵ William Rowlands,⁶ Melanie Watson,⁷ Kate Haralambos,⁸ Ryan Kenny,⁹ Dev Datta,¹⁰ Zosia Miedzybrodzka,¹¹ Christopher Byrne¹² and Joe Kai¹

Disclosure of interests

Full disclosure of interests: Completed ICMJE forms for all authors, including all related interests, are available in the tool kit on the NIHR journals Library report publication page at https://doi.org/10.3310/CTMD0148.

¹PRISM Research Group, Centre for Academic Primary Care, School of Medicine, University of Nottingham, Nottingham, UK

²Centre for Health Economics, University of York, York, UK

³Cardiovascular and Metabolism, Janssen Research and Development, High Wycombe, UK

⁴Primary Care, Population Sciences and Medical Education, University of Southampton, Southampton, UK

⁵Centre for Cardiovascular Genetics, Institute for Cardiovascular Science, University College London, London, UK

⁶Patient representative

⁷Wessex Clinical Genetics Service, University Hospital Southampton NHS Foundation Trust, Southampton, UK

⁸Familial Hypercholesterolaemia Service, University Hospital of Wales, Cardiff, UK

⁹Population Health Sciences Institute, Newcastle University, Newcastle upon Tyne, UK

¹⁰Lipid Unit, University Hospital Llandough, Penarth, UK

¹¹Division of Applied Medicine, University of Aberdeen, Aberdeen, UK

¹²Southampton National Institute for Health and Care Research Biomedical Research Centre, University Hospital Southampton NHS Foundation Trust, Southampton, UK

^{*}Corresponding author nadeem.gureshi@nottingham.ac.uk

Primary conflicts of interest: Nadeem Qureshi has received grants from the National Institute for Health and Care Research (NIHR) School for Primary Care Research (SPCR) and the Medical Research Council (MRC) (Newton-Ungku Omar Fund), has received honoraria from Amgen Inc. (Thousand Oaks, CA, USA) for lectures on familial hypercholesterolaemia, is a member of the board for the NIHR SPCR (2021 to present) and is a member of the Medical, Scientific and Research Committee of HEART UK. Bethan Woods reports membership of the NIHR Health Technology Assessment Clinical Evaluation and Trials Funding Committee (2020–24). Jo Leonardi-Bee has received a grant from the MRC to conduct research in the field of familial hypercholesterolaemia. Stephen Weng was part of an institution that received grants from the NIHR SPCR for research related to familial hypercholesterolaemia, has received consulting fees from the Academic Advisory Committee for RoadtoHealth Ltd (Hailsham, UK), has received honoraria and travel fees from Amgen Inc. for lectures on familial hypercholesterolaemia, was previously a committee member for the Medicines and Healthcare products Regulatory Agency Clinical Practice Research Datalink Independent Scientific Advisory Committee and is currently employed by Janssen R&D, High Wycombe, UK. Steve E Humphries has received support from the British Heart Foundation (PG 008/08); is the director of the UK Familial Hypercholesterolaemia Paediatric Register, which has received support from a grant from the International Atherosclerosis Society (Pfizer number 24052829); and is a medical director and minor shareholder of the University College London Business Ltd spinout company StoreGene (London, UK), which offers deoxyribonucleic acid (DNA) testing for individuals with familial hypercholesterolaemia. Kate Haralambos has received payment for PASS (PASS Software, Rijswijk, the Netherlands) clinical data entry and for extracting and processing data. Dev Datta has received personal honoraria from Daiichi Sankyo Company, Ltd (Tokyo, Japan), Pfizer Inc. (New York City, NY, USA), Ionis Pharmaceuticals, Inc. (Carlsbad, CA, USA), Amgen Inc., Akcea Therapeutics, Inc. (Ionis Pharmaceuticals, Inc.), Novartis AG (Basel, Switzerland) and Amryt Pharma plc (Dublin, Ireland), which are all manufacturers in the field of lipidology. Zosia Miedzybrodzka has received payment for arranging data extraction and for presenting data in Scottish health data Safe Haven (Grampian Data Safe Haven, University of Aberdeen) and has received funding from Amgen Inc. and Akcea Therapeutics, Inc. Joe Kai is a member of the NIHR SPCR board (2015 to present).

Published October 2023 DOI: 10.3310/CTMD0148

Plain language summary

Alternative cascade-testing protocols for identifying and managing patients with familial hypercholesterolaemia: systematic reviews, qualitative study and cost-effectiveness analysis

Health Technology Assessment 2023; Vol. 27: No. 16 DOI: 10.3310/CTMD0148

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain language summary

amilial hypercholesterolaemia is an inherited condition that causes raised cholesterol levels from birth and increases risk of heart disease if left untreated. After someone in a family is found to have familial hypercholesterolaemia (called an index case), their close relatives need to be contacted and checked to see if they have familial hypercholesterolaemia, using genetic or cholesterol testing. This is called 'cascade testing'. We planned to find the most cost-effective and acceptable way to do this.

The relatives could be contacted for testing by the index case (indirect approach), by a health-care professional (direct approach) or by a combination of both approaches. We found, based on looking at hospital records, that more relatives were tested if health-care professionals directly contacted relatives. In previous studies, slightly more relatives were tested for familial hypercholesterolaemia with a combination approach. Interviews with patients also suggested that the direct approach was the most effective, but the most acceptable and successful approach depends on family relationships: using one approach for some families and using both for other families.

Furthermore, by looking at the health-care records of large numbers of patients, we confirmed that people with a recorded diagnosis of familial hypercholesterolaemia in general practice records have a much higher risk of heart disease than the general population, and this was especially so for those with previous heart disease and/or raised cholesterols levels when diagnosed. However, one-quarter of new patients with familial hypercholesterolaemia recorded in their records were not treated within 2 years, with less than one-third reaching recommended cholesterol levels.

We used what we had learned to help us estimate the most cost-effective way to do cascade testing. This showed that if the health service directly contact all relatives simultaneously for further assessment, rather than the current approach whereby close (first-degree) relatives are contacted first, this was cost-effective and good value for money.

Health Technology Assessment

ISSN 1366-5278 (Print)

ISSN 2046-4924 (Online)

Impact factor: 3.6

Launched in 1997, *Health Technology Assessment* (HTA) has an impact factor of 3.6 and is ranked 32nd (out of 105 titles) in the 'Health Care Sciences & Services' category of the Clarivate 2021 Journal Citation Reports (Science Edition). It is also indexed by MEDLINE, CINAHL (EBSCO Information Services, Ipswich, MA, USA), Embase (Elsevier, Amsterdam, the Netherlands), NCBI Bookshelf, DOAJ, Europe PMC, the Cochrane Library (John Wiley & Sons, Inc., Hoboken, NJ, USA), INAHTA, the British Nursing Index (ProQuest LLC, Ann Arbor, MI, USA), Ulrichsweb™ (ProQuest LLC, Ann Arbor, MI, USA) and the Science Citation Index Expanded™ (Clarivate™, Philadelphia, PA, USA).

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full HTA archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hta.

Criteria for inclusion in the Health Technology Assessment journal

Reports are published in *Health Technology Assessment* (HTA) if (1) they have resulted from work for the HTA programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Reviews in *Health Technology Assessment* are termed 'systematic' when the account of the search appraisal and synthesis methods (to minimise biases and random errors) would, in theory, permit the replication of the review by others.

HTA programme

Health Technology Assessment (HTA) research is undertaken where some evidence already exists to show that a technology can be effective and this needs to be compared to the current standard intervention to see which works best. Research can evaluate any intervention used in the treatment, prevention or diagnosis of disease, provided the study outcomes lead to findings that have the potential to be of direct benefit to NHS patients. Technologies in this context mean any method used to promote health; prevent and treat disease; and improve rehabilitation or long-term care. They are not confined to new drugs and include any intervention used in the treatment, prevention or diagnosis of disease.

The journal is indexed in NHS Evidence via its abstracts included in MEDLINE and its Technology Assessment Reports inform National Institute for Health and Care Excellence (NICE) guidance. HTA research is also an important source of evidence for National Screening Committee (NSC) policy decisions.

This report

The research reported in this issue of the journal was funded by the HTA programme as project number 15/134/02. The contractual start date was in April 2017. The draft report began editorial review in May 2021 and was accepted for publication in June 2022. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HTA editors and publisher have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the draft document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health and Care Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, the HTA programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, the HTA programme or the Department of Health and Social Care.

Copyright © 2023 Qureshi *et al.* This work was produced by Qureshi *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This is an Open Access publication distributed under the terms of the Creative Commons Attribution CC BY 4.0 licence, which permits unrestricted use, distribution, reproduction and adaptation in any medium and for any purpose provided that it is properly attributed. See: https://creativecommons.org/licenses/by/4.0/. For attribution the title, original author(s), the publication source – NIHR Journals Library, and the DOI of the publication must be cited.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland, and final files produced by Newgen Digitalworks Pvt Ltd, Chennai, India (www.newgen.co).

NIHR Journals Library Editor-in-Chief

Dr Cat Chatfield Director of Health Services Research UK

NIHR Journals Library Editors

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HSDR, PGfAR, PHR journals) and Editorin-Chief of HSDR, PGfAR, PHR journals

Dr Peter Davidson Interim Chair of HTA and EME Editorial Board, Consultant Advisor, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Consultant in Public Health, Delta Public Health Consulting Ltd, UK

Ms Tara Lamont Senior Adviser, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Dr Catriona McDaid Reader in Trials, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK Professor Geoffrey Meads Emeritus Professor of Wellbeing Research, University of Winchester, UK

Professor Geoffrey Meads Emeritus Professor of Wellbeing Research, University of Winchester, UK

Professor James Raftery Professor of Health Technology Assessment, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Dr Rob Riemsma Consultant Advisor, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Professor Helen Roberts Professor of Child Health Research, Child and Adolescent Mental Health, Palliative Care and Paediatrics Unit, Population Policy and Practice Programme, UCL Great Ormond Street Institute of Child Health, London, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk